Author’s response to reviews

Title: Muslim women’s experiences of maternity services in the UK: Qualitative systematic review and thematic synthesis.

Authors:

Tasneema Firdous (tazzy_f@hotmail.co.uk)

Zoe Darwin (Z.J.Darwin@leeds.ac.uk)

Shaima Hassan (s.m.hassan@liverpool.ac.uk)

Version: 2 Date: 10 Feb 2020

Author’s response to reviews:

Dear Dr Rhona O’Connell

Thank you for giving us another opportunity to address the reviewers' comments.

Point 1: Please make sure the abstract contains only the following headings: Background, Methods, Results, Conclusions. The abstract now included all the highlighted headings.

Point 2: Please confirm whether the names used in your study are pseudonyms, and include a statement in your methods to state this. If they are real names, please change this to avoid publishing potentially-identifying information (of patients/participants). Yes the names included are pseudonyms. We have included a statement in the method section page 7 "Verbatim participants' quotations under pseudonyms were included" and we have included the terms 'pseudonym' following each participant name mentioned in the result section.

Point 3: Please remove the response to reviewers document from the end of your manuscript file as this is no longer needed in the publication process. This has been removed

Point 4: At this stage, please upload your manuscript as a single, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. A single manuscript is uploaded and figures and tables are uploaded as separate files.

Kind regards

Shaima M Hassan